Author's response to reviews

Title: Novel phenotypic variant in the MYH7 spectrum due to a stop-loss mutation in the C-terminal region: a case report

Authors:

Zsolt Bánfai (banfai.zsolt@pte.hu)
Kinga Hadzsiev (hadzsiev.kinga@pte.hu)
Endre Pál (pal.endre@pte.hu)
Katalin Komlósi (katalin.komlosi@unimedizin-mainz.de)
Márton Melegh (orygenet.titkarsag@pte.hu)
László Balikó (balikolaszlo@freemail.hu)
Béla Melegh (melegh.bela@pte.hu)

Version: 3 Date: 28 Aug 2017

Author’s response to reviews:

Dear Editor,

we revised the manuscript according to the Journal requirements.

- We added the required statement to the 'Consent to publish' section, which clarifies that the patient, described in the case report, consented the publication of medical data. (Photos of the patient were not presented in the case report.)

- A statement was inserted in the 'Case presentation' section as last sentence, stating that CARE guidelines were followed during the preparation of this manuscript.

- We attached to this submission the clean version of the manuscript only (with figures in separate files) as requested.

Thank you for the consideration of this revision.

I look forward to your decision.

Sincerely,
Bela Melegh, M.D., Ph.D., D.Sc.
Professor, Chairman
Department of Medical Genetics
University of Pécs
H-7624 Pecs, Szigeti 12.
Hungary
Tel: (+36)-72-536-427
Fax: (+36)-72-536-032
E-mail: melegh.bela@pte.hu